Author's response to reviews

Title: Improving access to care in low and middle-income countries: institutional factors related to enrollment efficiency and patient outcome in a cancer drug access program.

Authors:

Ebru Tekinturhan (Ebru.Tekinturhan@axiosint.com)
Etienne Audureau (etienne.audureau@yahoo.fr)
Marie-Pierre Tavolacci (Marie-Pierre.Tavolacci@chu-rouen.fr)
Patricia Garcia-Gonzalez (Pat.Garcia.Gonzalez@THEMAXFOUNDATION.ORG)
Joël Ladner (Joel.Ladner@chu-rouen.fr)
Joseph Saba (Joseph.Saba@axiosint.com)

Version: 2 Date: 2 May 2012

Author's response to reviews:

The following sentence has been added to the 'Methods / Data Collection' section:

"All data analyses were conducted under the authorization from Novartis and the Max Foundation and an Institutional Review Board (IRB) exemption was received from Axios IRB."